Author’s response to reviews

Title: A systematic review and integrative approach to decode the common molecular link between levodopa response and Parkinson’s disease

Authors:
Debleena Guin (Debleena.guin@igib.in)
Manish Mishra (manishkrmishra18@gmail.com)
Puneet Talwar (talwar.puneet@gmail.com)
Chitra Rawat (chitra.rawat@igib.in)
Suman Kushwaha (sumankushwaha@gmail.com)
Shrikant Kukreti (shrikant.kukreti6@gmail.com)
Ritushree Kukreti (ritus@igib.in)

Version: 2 Date: 23 Aug 2017

Author’s response to reviews:
1. Email of all authors
The change has been incorporated in the title page of the uploaded revised manuscript

2. Corresponding author details
The change has been incorporated in the title page of the uploaded revised manuscript

3. Methods
The change has been incorporated in the uploaded revised manuscript

4. Additional files list
The change has been incorporated in the uploaded revised manuscript.
5. Declarations
The change has been incorporated in the uploaded revised manuscript.

6. Acknowledgements
The change has been incorporated in the uploaded revised manuscript.

7. Table and figure legends
The change has been incorporated in the uploaded revised manuscript.

8. Supplementary material
The point-by-point response from the supplementary material has been excluded.

9. Clean version
The clean version of the manuscript has been uploaded.